# Ondansetron and metoclopramide as second-line antiemetics in women with nausea and vomiting in pregnancy: the EMPOWER pilot factorial RCT

Stephen Robson,<sup>1\*</sup> Catherine McParlin,<sup>2</sup>
Helen Mossop,<sup>1</sup> Mabel Lie,<sup>1</sup>
Cristina Fernandez-Garcia,<sup>1</sup> Denise Howel,<sup>1</sup>
Ruth Graham,<sup>3</sup> Laura Ternent,<sup>1</sup> Alison Steel,<sup>4</sup>
Nicola Goudie,<sup>4</sup> Afnan Nadeem,<sup>4</sup> Julia Phillipson,<sup>4</sup>
Manjeet Shehmar,<sup>5</sup> Nigel Simpson,<sup>6</sup> Derek Tuffnell,<sup>7</sup>
Ian Campbell,<sup>8</sup> Rew Williams,<sup>9</sup> Margaret E O'Hara,<sup>9</sup>
Elaine McColl<sup>1</sup> and Catherine Nelson-Piercy<sup>10</sup>

<sup>&</sup>lt;sup>1</sup>Population Health Sciences Institute, Faculty of Medical Sciences, Newcastle University, Newcastle upon Tyne, UK

<sup>&</sup>lt;sup>2</sup>Department of Nursing, Midwifery and Health, Faculty of Health and Life Sciences, Northumbria University, Newcastle upon Tyne, UK

<sup>&</sup>lt;sup>3</sup>School of Geography, Politics and Sociology, Newcastle University, Newcastle upon Tyne, UK

<sup>&</sup>lt;sup>4</sup>Newcastle Clinical Trials Unit, Newcastle University, Newcastle upon Tyne, UK

<sup>&</sup>lt;sup>5</sup>Gynaecology Secretaries Department, Birmingham Women's and Children's NHS Foundation Trust, Birmingham, UK

<sup>&</sup>lt;sup>6</sup>Leeds Institute of Medical Research, Department of Women's and Children's Health, School of Medicine, University of Leeds, Leeds, UK

<sup>&</sup>lt;sup>7</sup>Department of Obstetrics, Bradford Teaching Hospitals NHS Foundation Trust, Bradford, UK

<sup>&</sup>lt;sup>8</sup>Pharmacy Department, Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle upon Tyne, UK

<sup>&</sup>lt;sup>9</sup>Patient representative

<sup>&</sup>lt;sup>10</sup>Department of Obstetric Medicine, Guy's and St Thomas' Foundation Trust, London, UK

<sup>\*</sup>Corresponding author s.c.robson@ncl.ac.uk

Declared competing interests of authors: Stephen Robson reports that he was a member of the National Institute for Health Research (NIHR)/Medical Research Council Efficacy and Mechanism Evaluation Funding Committee (2012–15). Catherine Nelson-Piercy reports personal fees from Alliance Pharma (Chippenham, UK), UCB Biopharma SPRL (Brussels, Belgium), Sanofi SA (Paris, France) and Janssen: Pharmaceutical Companies of Johnson & Johnson (Beerse, Belgium) outside the submitted work. Denise Howel reports being a member of the NIHR Health Services and Delivery Research Commissioning Board (January 2012 to November 2015) and being a member of the NIHR Programme Grants for Applied Research (PGfAR) subpanel (from February 2016 to February 2020). Manjeet Shehmar reports personal fees from Alliance Healthcare (Chessington, UK), outside the submitted work, and Secretary Association Early Pregnancy Units UK (London, UK). Manjeet Shehmar also reports being lead author of the Royal College of Obstetricians and Gynaecologists Green-top Guideline No. 69, 2016. Margaret O'Hara reports personal fees from Alliance Pharma outside the submitted work. Elaine McColl reports membership of the NIHR PGfAR editorial board (2013–16), NIHR PGfAR subpanel (2008–16), NIHR Clinical Trials Unit Standing Advisory Committee (2015–16) and NIHR Journals Library Editorial Group (2014–15).

Published November 2021

DOI: 10.3310/hta25630

# Plain English summary

The EMPOWER pilot factorial RCT

Health Technology Assessment 2021; Vol. 25: No. 63

DOI: 10.3310/hta25630

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

ausea and vomiting in pregnancy cause physical and emotional distress, and up to 30% of affected women require medical treatment. Guidelines on the use of anti-sickness drugs exist, but evidence is limited about which drugs work the best.

The EMPOWER (EMesis in Pregnancy – Ondansetron With mEtoClopRamide) trial aimed to compare the clinical effectiveness and cost-effectiveness of two anti-sickness drugs [metoclopramide (metoclopramide hydrochloride, Actavis UK Ltd, Barnstable, UK; IV Ratiopharm GmbH, Ulm, Germany) and ondansetron (ondansetron hydrochloride dehydrate, Wockhardt UK Ltd, Wrexham, UK; IV Hameln Pharma plus GmbH, Hameln)] for the treatment of nausea and vomiting in pregnancy. Women who were < 17 weeks pregnant with severe nausea and vomiting who attended hospital because their first anti-sickness drug had failed to improve their symptoms were asked to take part in the trial. Participants received fluids and, with consent, were randomly allocated to one of four groups: (1) metoclopramide and dummy ondansetron, (2) ondansetron and dummy metoclopramide, (3) metoclopramide and ondansetron or (4) double dummy. Trial drugs were administered into a vein and then by tablet for 10 days. On advice from sufferers, the trial focused on treatment failure, but other outcomes, including drug side effects, costs and pregnancy outcome, were collected.

The trial was unable to recruit enough women and, therefore, did not progress. Nearly 600 women at 11 hospitals were screened, of whom 122 (21%) were eligible and 33 were recruited. The main reason for ineligibility (68%) was prior use of trial drug (mostly ondansetron). Overall, 15 out of 30 evaluable women experienced treatment failure.

Interviews with 21 women who were approached about the trial and 22 research staff identified complex hurdles to and enablers of recruitment. The main hurdles were the requirements of the study protocol in relation to guidelines on anti-sickness drugs and the diversity of pathways to care. The role of research staff was a key enabler.

The trial was too small to draw useful conclusions and it highlights the challenges of conducting complex studies on sick pregnant women. Subsequent concerns about the safety of ondansetron highlight the need for further studies to help inform women and the NHS about the best care for nausea and vomiting in pregnancy.

## **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

### **HTA** programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 16/15/03. The contractual start date was in September 2017. The draft report began editorial review in September 2020 and was accepted for publication in February 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

Copyright © 2021 Robson *et al.* This work was produced by Robson *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk